

JGR Biogeosciences

RESEARCH ARTICLE

10.1029/2022JG007137

Key Points:

- Current global lake methane release estimated to account for 11% of the global total natural source emissions
- Future global lake emissions are estimated to increase 58%–86% under the severe climate warming scenario RCP8.5
- Under warming conditions, methanogenesis will increase, but enhanced methane oxidation in water column can dampen the net methane emission

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

Q. Zhuang, qzhuang@purdue.edu

Citation:

Zhuang, Q., Guo, M., Melack, J. M., Lan, X., Tan, Z., Oh, Y., & Leung, L. R. (2023). Current and future global lake methane emissions: A process-based modeling analysis. *Journal of Geophysical Research: Biogeosciences*, *128*, e2022IG007137. https://doi. org/10.1029/2022JG007137

Received 24 AUG 2022 Accepted 8 MAR 2023

Author Contributions:

Conceptualization: Qianlai Zhuang Data curation: Mingyang Guo Formal analysis: Mingyang Guo, Xin Lan Funding acquisition: Qianlai Zhuang Investigation: Mingyang Guo Methodology: John M. Melack Supervision: Qianlai Zhuang Validation: Mingyang Guo Visualization: Mingyang Guo Writing – original draft: Mingyang Guo, Xin Lan

© 2023 The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

Current and Future Global Lake Methane Emissions: A Process-Based Modeling Analysis

Qianlai Zhuang^{1,2}, Mingyang Guo¹, John M. Melack³, Xin Lan^{4,5}, Zeli Tan⁶, Youmi Oh⁴, and L. Ruby Leung⁶

¹Department of Earth, Atmospheric and Planetary Sciences, Purdue University, West Lafayette, IN, USA, ²Purdue Climate Change Research Center, West Lafayette, IN, USA, ³Bren School of Environmental Science and Management and Earth Research Institute, University of California, Santa Barbara, CA, USA, ⁴Global Monitoring Laboratory, National Oceanic and Atmospheric Administration, Boulder, CO, USA, ⁵Cooperative Institute for Research in Environmental Sciences, University of Colorado Boulder, Boulder, CO, USA, ⁶Pacific Northwest National Laboratory, Richland, WA, USA

Abstract Freshwater ecosystem contributions to the global methane budget remains the most uncertain among natural sources. With warming and accompanying carbon release from thawed permafrost and thermokarst lake expansion, the increase of methane emissions could be large. However, the impact and relative importance of various factors related to warming remain uncertain. Based on diverse lake characteristics incorporated in modeling and observational data, we calibrate and verify a lake biogeochemistry model. The model is then applied to estimate global lake methane emissions and examine the impacts of temperature increase for the first and the last decades of the 21st century under different climate scenarios. We find that current emissions are 24.0 ± 8.4 Tg CH₄ yr⁻¹ from lakes larger than 0.1 km², accounting for 11% of the global total natural source as estimated based on atmospheric inversion. Future projections under the RCP8.5 scenario suggest a 58%–86% growth in emissions from lakes. Our model sensitivity analysis indicates that additional carbon substrates from thawing permafrost may enhance methane production under warming in the Arctic. Warming enhanced methane oxidation in lake water can be an effective sink to reduce the net release from global lakes.

Plain Language Summary Methane is the second most important greenhouse gas after carbon dioxide but with much stronger warming potential. Providing accurate estimates and future projections of methane emissions are challenging but essential for supporting climate mitigation strategies. Freshwater ecosystems including lakes and reservoirs contribute large uncertainty in the quantification of global methane sources. In this study, with an improved lake modeling approach, we estimate global lake methane emissions and examine the impacts of temperature increase, permafrost thaw, and thermokarst lake area changes in the future. Our modeling suggests that the current magnitude and the future increase of global lake methane emissions are lower than previously suggested. Under warming conditions, the increase of methane oxidation reduces the net lake methane emissions.

1. Introduction

Over the past two decades, freshwater ecosystems including lakes, ponds, and reservoirs have received increasing attention as observations indicate that they are a disproportionately large methane source compared to their surface areas (Saunois et al., 2016, 2020). Recent bottom-up estimates of global freshwater methane emissions (Johnson et al., 2022; Saunois et al., 2020) are 2–4 times larger than top-down estimates of all natural sources excluding wetlands (Saunois et al., 2016, 2020). For the region north of 50°N with the largest lake coverage and climate sensitivity, statistical upscaling (Walter et al., 2007; Wik et al., 2016) and process-based modeling (Tan & Zhuang, 2015) suggests that over 60% of the total natural methane emission is from lakes, which disagrees with some previous estimates suggesting a dominant contribution from wetlands (Fung et al., 1991; Melton et al., 2013). These disparities might be central to explaining the differences between top-down emission estimates based on atmospheric measurements and bottom-up estimated net emissions (Ronsentreter et al., 2021; Saunois et al., 2016, 2020). Both top-down and bottom-up results can be biased due to (a) low temporal resolution of most observations of methane fluxes; (b) over-representation or under-representation of lakes from certain regions and of certain properties, such as the bias toward more frequent sampling of small northern lakes



Writing – review & editing: Qianlai Zhuang, Mingyang Guo, John M. Melack, Xin Lan, Zeli Tan, Youmi Oh, L. Ruby Leung compared to large or tropical lakes; and (c) disagreements in lake abundance and surface areas because of different mapping approaches.

Climate warming is likely to increase methane production from lakes and reservoirs, especially in the circumpolar permafrost region where gradual thaw can expose previously frozen carbon to microbial decomposition and abrupt thaw can accelerate the formation and expansion of thermokarst lakes (Schuur et al., 2009, 2015; Walter Anthony et al., 2016, 2018). Measurements of 37 permafrost-affected lakes over the past 60 years have shown 0%–4,000% higher methane ebullitive emissions in the expansion zones than in open water (Walter Anthony et al., 2016). Lakes formed through glacial and periglacial processes during the last deglaciation are estimated to represent around 78% of northern lake surface area (Wik et al., 2016). These lakes were projected to increase their emissions by 5 and 26 Tg CH₄ yr⁻¹ due to newly thawed carbon within this century under the representative concentration pathway (RCP) 2.6 and RCP8.5 scenarios, respectively (Von Deimling et al., 2015). However, the contributions of arctic lakes to the global methane budget in a warmer future are still uncertain.

Here, we present a process-based modeling analysis of global lake methane emissions for the first and the last decades of the 21st century under different climate scenarios. We adapt a one-dimensional (1-D) process-based lake biogeochemical model (Tan et al., 2015, 2017) by modifying the methane module for various climate zones and lake and sediment properties. The model calibration and validation data sets are organized based on the temporal and spatial coverages, frequency of observations, and measurement techniques. We use the Hydro-LAKES database (Messager et al., 2016) which includes all lakes and reservoirs larger than 0.1 km² as the basis for our simulations. We discuss reasons behind the different estimates by statistical upscaling and process-based modeling. We then investigate the significance of different climate-induced factors, including direct warming, permafrost sediment organic carbon (SOC) input, and areal changes in thermokarst lakes. We aim to reduce the uncertainties in freshwater methane emissions and improve our understanding of how lake ecosystems respond to climate change.

2. Methods

2.1. Model Configuration

The Arctic Lake Biogeochemical Model (ALBM) is a 1-D process-based model that simulates water and sediment thermal conditions and carbon and nutrient dynamics (Guo, Zhuang, Yao, Golub, Leung, & Tan, 2021; Tan et al., 2015, 2017; Figure S1 in Supporting Information S1). The model has been applied to lakes in various regions apart from the arctic (Guo et al., 2020; Guseva et al., 2020; Tan et al., 2018). The thermal module simulates radiative heating, turbulent heat transfer, ice phenology, heat diffusion and convection within water columns, sediment-water interface heat exchange, and sediment heat diffusion. More details can be found in Guo, Zhuang, Yao, Golub, Leung, Pierson, et al. (2021) and Guo, Zhuang, Yao, Golub, Leung, and Tan (2021) who evaluated the module against a variety of lakes, improved the algorithms through model intercomparisons, and calibrated the parameters to suit global-scale simulations. Here for simplicity, we briefly describe the algorithms for methane related processes which are the foci of this study. Detailed biogeochemical processes, including phytoplankton and other microbial activities and oxygen dynamics can be found in Tan et al. (2015, 2017). Full parameter definitions of the methane module are listed in Table 1. Some of the parameter values vary by lake type as explained in the next section.

Methane production (P) in the lake sediment is calculated as

$$P = R_c C_{\text{labile}} P Q_{10}^{\frac{T_s - T_{p0}}{10}}$$
(1)

where R_c is the carbon decomposition rate (s⁻¹), C_{labile} is the labile carbon in upper 0.3 m of sediment (kg C m⁻²), PQ_{10} is the Q10 factor representing the sensitivity to temperature, T_s is the sediment temperature (C), and T_{p0} is the reference temperature (C). $C_{\text{labile}} = C_{\text{pool}} \alpha_{\text{labile}}$, where C_{pool} is the total organic carbon and α_{labile} is the fraction of labile carbon. The carbon pool is considered constant over the simulation period. Methane is transported through the water column by diffusion and ebullition. The concentration in the water column (C_{CH_d}) is calculated as

$$\frac{\partial C_{\text{CH}_4}}{\partial t} = \frac{\partial}{\partial z} \left(D_{\text{CH}_4} \frac{\partial C_{\text{CH}_4}}{\partial z} \right) - O + E_{\text{CH}_4}$$
(2)



Table 1

Model Parameters in the Methane Module

Symbol	Lake type	Value	Units	Units References	
α_{labile}	Pan-Arctic permafrost-affected	20	%	Schädel et al. (2014)	
	Oligo/mesotrophic, shallow warm water	15		Ostapenia et al. (2009) and Klump et al. (1989)	
	Eutrophic	22		Ostapenia et al. (2009)	
	Hypereutrophic	45		Keaveney et al. (2020)	
T_{p0}	-	3.5	С	Tan et al. (2015)	
T_{o0}	-	-5.5	С	Zhuang et al. (2004)	
k_{O_2}	-	1.4×10^4	$\mu mol m^{-3}$	Segers (1998) and van Bodegom et al. (2001)	
Symbol	Value range	Units	Reference		
R _c	$[1.342 \times 10^{-10}, 1.342 \times 10^{-9}]$	s ⁻¹	Kessler et al. (2012)		
PQ_{10}	[1.7, 16]	-	Walter and Heimann (2000)		
O_{\max}	$[3.7 \times 10^{-3}, 1.6]$	$\mu mol m^{-3} s^{-1}$	Liikanen et al. (2002), Lofton et al. (2014), and Remsen et al. (1989)		
OQ_{10}	[1.4, 3.5]	-	Tang and Zhuang (2009)		
$k_{ m CH_4}$	$[1.0 \times 10^{-3}, 6.62 \times 10^{-4}]$	$\mu mol m^{-3}$	Segers (1998)		

Note. Calibrated parameters are listed with ranges.

where D_{CH_4} is eddy diffusivity or entrainment, *O* is aerobic methane oxidation, and E_{CH_4} is the gas exchange with bubbles. Methane exchanges at the sediment-water and water-air interfaces are added to the bottom and top boundaries, respectively. Methane oxidation (*O*) in the water column is a function of water temperature and methane concentrations

$$O = O_{\max} O Q_{10}^{\frac{T_w - T_{00}}{10}} \frac{C_{O_2}}{k_{O_2} + C_{O_2}} \frac{C_{CH_4}}{k_{CH_4} + C_{CH_4}}$$
(3)

where O_{max} is the oxidation potential (μ mol m⁻³ s⁻¹), OQ_{10} is the Q10 factor, T_w is the water temperature (C), T_{o0} is the reference temperature (C), C_{O_2} and C_{CH_4} are gas concentrations (μ mol m⁻³), and k_{O_2} and k_{CH_4} are Michaelis-Menten constants (μ mol m⁻³). The gas diffusive flux (F) at the water-air interface is calculated as

$$F = k_{600}(C_w - C_{eq})$$
(4)

where k_{600} is gas transfer velocity normalized to CO₂ at 20°C, C_w is near-surface concentration of dissolved CH₄, and C_{eq} is near-surface concentration in equilibrium with the atmosphere. The calculation of gas transfer velocity follows Heiskanen et al. (2014): $k_{600} = \sqrt{(c_1 U)^2 + (c_2 w_*)^2} S c^{-0.5}$, where c_1 and c_2 are empirical constants, U is the near-surface wind speed (m s⁻¹), $w_* = (\beta z_{AML})^{1/3}$, β is buoyancy flux, z_{AML} is the mixing layer depth, and Scis the Schmidt number. The governing equations of bubble transfer in the water column are adapted from ocean models (Liang et al., 2011; Woolf & Thorpe, 1991). The gas concentration in bubbles of certain size at certain depth and time step is calculated by a continuity equation in which the impacts of buoyant rising, gas exchange with ambient water, and bubble expansion are included (Liang et al., 2011). Gases in bubbles are assumed to release instantly to the atmosphere upon reaching the lake surface. Detailed formulation can be found in Tan et al. (2015).

2.2. Observational Data on Methane Emissions

We filtered flux measurement data to avoid short observation periods, inadequate spatial coverage, and under-represented seasonal variability. For the tropical regions with shallow lakes that experience large variations in water level and methane fluxes throughout the year, observations that cover at least 1 year and a range of water levels were selected (Amaral et al., 2020; Attermeyer et al., 2016; Marani & Alvalá, 2007). Therefore, observations that cover the hydrological cycle for at least 1 year were selected. For all other regions, a minimum of



3 months, at least one ice-free season and a sampling frequency that exceeded 1 day per month were required for lake observations. For lakes larger than 10 km², data from at least two methods, such as floating chambers and bubble traps, were required. If the boundary layer method was used to calculate diffusive fluxes, studies using wind speed measured simultaneously onsite instead of taken from more distant meteorological stations were selected. As a result, out of the 838 lakes and reservoirs with published data, 60, varying in size and climate zone (Figure S1 and File S1 in Supporting Information S1), were used for model calibration and validation. Fifty-five have observed diffusive emissions and 45 include ebullition observations. The physical properties (coordinates, surface area, and depth) and water and sediment properties (trophic status, thermokarst or not, yedoma or not) of the lakes, information of the measurements (time period, number of measurements, and technique), and the mean daily/annual fluxes are included in (File S1 in Supporting Information S1). Here, lakes and reservoirs were not treated as separate categories due to the lack of sufficient reservoir observations. The selected sites were first classified by climate zone and then by lake water nutrient level and sediment type. For tropical and subtropical regions, due to the low number of observation sites, they were divided into shallow warm water ($30^{\circ}N-30^{\circ}S$, mean depth <10 m) and other lakes (Table S1 in Supporting Information S1).

2.3. Model Calibration and Validation

The calibration of parameters in the thermal module was done in a previous study (Guo, Zhuang, Yao, Golub, Leung, & Tan, 2021) and thus, we calibrated the model for methane fluxes here. Observations with both diffusive and ebullitive emissions were used for parameter calibration and the remainder of observations for validation. For each site, the model was run over the observation period and parameters were calibrated against the mean daily fluxes during that period. For the lakes with only mean annual fluxes provided, we divided the values by the annual mean modeled length of ice-free days to obtain the mean daily fluxes.

The calibration was carried out using a Bayesian approach in which a uniform distribution was used for all parameters, and Monte-Carlo-based calibrations were conducted more than once by sampling from the posterior distribution obtained from the previous round (Guo, Zhuang, Yao, Golub, Leung, & Tan, 2021; Tang & Zhuang, 2009). The number of the sampling rounds was case-sensitive and decided by the model performance. The size of the perturbed parameter ensemble (PPE) equals the number of parameters multiplied by 1,000 in each round. The root-mean-square errors (RMSEs) of the simulated emissions during the same periods as the observations were calculated.

Lakes in the same category were calibrated together, meaning that the optimal parameters with the minimum mean RMSEs of all lakes in that category were selected. The calibration stops when RMSE is <10%. We did not calibrate the model against the time series of measurements because: (a) some observational data were only daily or annual means; (b) the temporal resolutions vary among sites, and therefore, using the time series will bias the model toward the observation-dense sites; and (c) the focus of this study is the annual mean emissions because long-term continuous measurements are very rare.

2.4. Global Lake Classification and Simulation

We used the HydroLAKES database (Messager et al., 2016) for global lake and reservoir mapping with information on coordinates, areas, and depths, which remained constant through all the simulation periods. Due to lack of bathymetric data, the horizontal cross-sectional area is assumed to decrease linearly for all lakes. Lakes were classified into various categories for model calibration (Figure S2 in Supporting Information S1), and the calibrated parameter values were applied to all lakes for the same category. The global SOC content is approximated from the global soil organic carbon map (GSOCmap, http://54.229.242.119/GSOCmap/). For the pan-Arctic region, the yedoma map was based on the Ice-Rich Yedoma Permafrost (IRYP, Strauss et al., 2016) data set, the permafrost map from Brown et al. (1997), and the thermokarst lake distribution from Olefeldt et al. (2016). For northern temperate lakes, the trophic status was determined by a combination of the Lake Water Quality (LWQ) data set that includes a trophic state index, chlorophyll concentrations (Filazzola et al., 2020), and the Global River Chemistry Database (GLORICH; Hartmann et al., 2014) that includes total nitrogen and phosphorus concentrations. Chlorophyll and solute concentrations were converted into trophic state indices based on the OECD (OECD, 1992) classification scheme. Lakes not included in these data sets were classified as oligotrophic.

Simulations were run from 2004 to 2006 (hereafter refer to as the 2000s), and 2094 to 2096 (hereafter the 2090s) under RCP8.5 scenarios with a spinup period of 10 years using climate forcing data from 1994 to 2003 and



2084 to 2093, respectively. The required meteorological forcing data include surface air temperature, minimum and maximum surface air temperature, surface air pressure, relative humidity, 10-m wind speed, precipitation, snowfall, surface downward shortwave radiation, and surface downward longwave radiation. Inflow and outflow observations were not available at the global scale; therefore, we assumed that water levels are static. The forcing data for present and future scenarios were supplied by the ISIMIP2a bias-corrected climate input data by HadGEM2-ES and GFDL-ESM2M climate models (Frieler et al., 2017; Lange, 2019), which represent one of the severest and the mildest warming scenarios under RCP8.5, respectively, among climate models. The data are at $0.5^{\circ} \times 0.5^{\circ}$ grids and daily time step from 1979 to 2016 and 2006 to 2099, respectively.

The ALBM is a site-level model, meaning that lakes are simulated individually. Given the large number of lakes globally (\approx 1.4 million), we reduced the total computational time by avoiding repetitive simulations of similar lakes. Specifically, lakes of the same type and similar depths and shapes (= $\sqrt{\text{area}/\text{depth}}$) within each $0.5^{\circ} \times 0.5^{\circ}$, the same resolution as the climate forcing data, were considered as a cohort and simulated as such. The mean coordinates, depth, surface area, and elevation were used as the simulation input for each cohort, and modeled fluxes were considered uniform for all lakes in the same cohort. Depths were binned with an interval of 1 m within (0,10] m and 10 m within (10,100] m. Shape factors are binned by (0, 50, 100, 200, 300, 400, 500, 600, 800, 1,000]. This manipulation helped to reduce about 27% of the total simulated lakes, especially in regions dense with small lakes (Figure S3 in Supporting Information S1).

The site uncertainty is quantified as the normalized root-mean-square-error NRMSE = $RMSE/\overline{P}$, *P* is model prediction. We assume the NRMSE values for calibrated lakes are constants for each lake type, which are then used to generate regional emission uncertainties based on lake types and their simulated emissions.

2.5. Experiments on Permafrost Dynamics Impacts

The large stocks of frozen organic carbon in permafrost soils might be exposed to microbial decomposition under warming-induced thaw. To estimate this impact on methane emissions from arctic lakes, we estimate lake area changes in yedoma and nonyedoma continuous and discontinuous permafrost regions based on the predicted permafrost maps under the RCP4.5 and RCP8.5. The permafrost maps were estimated based on the simulated soil temperature by MRI-CGCM3 (Yukimoto et al., 2012). We followed the method in Koven et al. (2013) to estimate both the historical and future extent of lake area. With the lake area having permafrost reduced by about 96% under the RCP8.5 (Table S4 in Supporting Information S1), lake methanogenesis might be significantly fueled by old organic carbon. Without regional-scale measurements or modeling results of the amount of permafrost organic carbon adding to lakes, we cannot explicitly simulate the impacts on lake methane emissions. Instead, we conducted a sensitivity test by doubling the lake SOC concentration under the RCP8.5 using the HadGEM2-ES climate model output. We acknowledge this is a hypothetical modeling experiment, although previous modeling and observational-based inventory studies all suggest increasing carbon availability due to thawing permafrost under future warming scenarios (Hugelius et al., 2020; Koven et al., 2015).

Thermokarst lakes in the carbon-rich yedoma and western Siberia regions have high methane emissions (Serikova et al., 2019; Walter Anthony et al., 2012). These lakes often persist for short periods compared to other lakes, and therefore, are more variable under climate change (van Huisstden et al., 2011). Abrupt permafrost thaw can cause rapid lake formation and large-scale gradual thaw will facilitate expansion and deepening of the lake (Turetsky et al., 2020; Walter Anthony et al., 2016). Alternatively, the changing hydrology may promote lateral breaching and thus, drainage of the lakes (Jones et al., 2011; Nitze et al., 2020). Both expansion and drainage have been observed across the pan-Arctic permafrost region whereas the dominant processes, magnitudes, and time scales are heterogeneous (Table S5 in Supporting Information S1). To evaluate the impacts on methane emissions, we assumed a change in the total thermokarst lake area by $\pm 30\%$ by the end of the century, which covers most of the observations over 30–60 years and is also consistent with Wik et al. (2016) for comparison purpose.

3. Results and Discussion

3.1. Model Calibration and Validation

Overall, the model showed good performance with NRMSEs of 0.5 (0.66), 1.07 (0.27), and 1.28 (0.27) mmol $m^{-2} day^{-1}$ in calibration (validation) and MBEs (mean bias error) of 0.26 (0.11), -0.15 (-0.14),



10.1029/2022JG007137



Figure 1. Biases for lakes in calibration and validation for total methane fluxes (a, b), diffusive fluxes (c, d), and ebullitive fluxes (e, f). Lakes with both observed diffusive and ebullitive fluxes were used for calibration and those with only one observed pathway or only total fluxes were used for validation.

and 0.11 (-0.06) mmol m⁻² day⁻¹ for simulated diffusive, ebullitive, and total emissions, respectively (Figure 1). The metrics for each lake are listed in Tables S2 and S3 in Supporting Information S1. The model tends to produce lower diffusive fluxes than the observations. The calculation of vertical mixing and gas transfer velocity at water-atmosphere interface may contribute to the underestimation. Furthermore, floating chambers may occasionally capture bubbles, leading to overestimated diffusive fluxes.

Table 2

Global Total Methane Emissions $(E_{CH_4} (Tg yr^{-1}))$ in Different Climate							
Zones Under Present and Future Climate Scenarios and the Percentage							
Increases (ΔE_{CH_4} (%)) With Respect to the 2000s							

Scenario	Value	Pan-arctic	Temperate	Tropics	All
Present	$E_{\rm CH_4}$	7.4	9.7	6.9	24.0
2090sHadGEM2-ES	$E_{\rm CH_4}$	23.5	12.8	8.3	44.6
	ΔE_{CH_4}	217.6	31.3	21.0	85.8
2090sGFDL-ESM2M	$E_{\rm CH_4}$	19.1	11.6	7.1	37.8
	ΔE_{CH_4}	166.1	19.4	2.6	57.5

3.2. Current Emissions and Their Spatial Distribution

The simulated mean annual methane emissions from global lakes and reservoirs larger than 0.1 km² are 24.0 ± 8.4 Tg CH₄ yr⁻¹, 2.6 ± 1.2 Tg CH₄ yr⁻¹ is from diffusion and 21.4 ± 7.2 Tg CH₄ yr⁻¹ is from ebullition. The temperate zone (23.5°–50°N, 23.5°–60°S) is the largest source contributing 9.7 ± 3.4 Tg CH₄ yr⁻¹, followed by 7.4 ± 2.6 Tg CH₄ yr⁻¹ and 6.9 ± 2.4 Tg CH₄ yr⁻¹ for the pan-Arctic (>50°N) and the tropics (23.5°N–23.5°S), respectively (Table 2 and Figure 2). The emission per unit lake area has different latitudinal distributions with 16.9 g CH₄ m⁻² yr⁻¹ for the tropics, 8.0 g CH₄ m⁻² yr⁻¹ for the temperate region, and 5.6 g CH₄ m⁻² yr⁻¹ for the distribution patterns in the total emission and areal fluxes are related to the right-skewed distribution of lake surface areas across the globe (Figure 2b). The pan-Arctic total lake surface area (1.3 × 10⁶ km²)





Figure 2. Mean annual lake methane fluxes weighted by lake area (a). Latitudinal distributions of total number and surface area of lakes in HydroLAKES with 1° intervals (b), annual total methane emissions (c), and annual methane fluxes averaged by lake area (d). The smoothed curves are fitted by polynomial regression and shades are the uncertainty ranges.

is larger than the areas in temperate $(1.2 \times 10^6 \text{ km}^2)$ and tropical regions $(0.4 \times 10^6 \text{ km}^2)$, and thus the total emissions from the pan-Arctic lakes are high even with the lower areal fluxes.

Our estimate is on the lower end of the 16.9–67.4 Tg yr⁻¹ (first and third quartiles, median = 32.1 Tg yr⁻¹) of the estimate by Rosentreter et al. (2021) for lakes larger than 0.1 km² and all reservoirs; the estimated emission from reservoirs smaller than 0.1 km² was only 0.4 Tg yr⁻¹. The estimated total for the pan-Arctic is about 55% lower than 16.5 \pm 9.2 (mean \pm 0.5 interquartile) Tg yr⁻¹ estimated by Wik et al. (2016) which includes all lakes larger than 0.002 km². Lakes smaller than 0.1 km², although minor in the total surface area, were identified as a disproportionally major, but especially uncertain, source in Rosentreter et al. (2021).

Both Wik et al. (2016) and Rosentreter et al. (2021) used the global water bodies (GLOWABO) data set (Verpoorter et al., 2014) for lakes larger than 0.002 km², and Rosentreter et al. (2021) included smaller lakes by statistically extrapolating the abundance and area. Messager et al. (2016) developed the HydroLAKES data set by manually correcting for fluvial features, and reducing the uncertainties caused by double counting small lakes and wetlands. For lakes larger than 0.1 km², GLOWABO reports higher lake abundance than HydroLAKES especially in the 0.1–1 km² size classes with 204% more lakes, and the discrepancy decreases with increasing size class. If we



assume the same discrepancy, these additional lakes would add 12.7 Tg yr⁻¹ (median, 5.3–34.4 Tg yr⁻¹, first and third quartiles) more emissions, leading to total 36.7 Tg yr⁻¹ from all lakes.

Reservoirs are significant methane sources due, in part, to flooding of terrestrial organic carbon, enhanced ebulition during drawdown, and nutrients from anthropogenic activities (Deemer et al., 2016). In our study, although reservoirs and natural lakes were not differentiated, six reservoirs were included in the calibration and validation. The model overestimated the total fluxes by $0.1-0.5 \text{ mmol m}^{-2} \text{ day}^{-1}$ for four of the sites and underestimated the fluxes by $0-0.2 \text{ mmol m}^{-2} \text{ day}^{-1}$ for the other two. HydroLAKES includes 6,687 reservoirs covering $2.49 \times 10^5 \text{ km}^2$, about 9% of the total surface area of lakes and reservoirs. The modeled reservoir emissions are $2.1 \text{ Tg CH}_4 \text{ yr}^{-1}$, $0.4 \text{ Tg CH}_4 \text{ yr}^{-1}$ by diffusion and $1.7 \text{ Tg CH}_4 \text{ yr}^{-1}$ by ebullition, contributing ~9% of the total emission. Johnson et al. (2021) estimated total reservoir emissions of $10.1 \text{ Tg CH}_4 \text{ yr}^{-1}$ (7.2–12.9 Tg CH₄ yr⁻¹) using a statistical upscaling approach, almost five times higher than our estimates, based on a data set of 42,000 reservoirs summing up to $2.97 \times 10^5 \text{ km}^2$. Johnson et al. (2021) also used HydroLAKES for areas and abundances but included more small reservoirs with a total area about 20% greater than that in our modeling. Furthermore, to provide estimates of annual fluxes when the observational data did not include all seasons, Johnson et al. (2021) used two statistically fitted equations from previous studies to relate diffusive and ebullitive emissions to surface air temperature, respectively. Applying the equations developed for a few specific sites to the global lakes can lead to biases. Also, the relationship between air and water temperature is not well represented.

Our estimates of global lake and reservoir emissions are consistent with current global top-down estimates of natural emissions ranging from 21 to 47 Tg CH₄ yr⁻¹ excluding wetlands (Saunois et al., 2020). The top-down emission estimates optimize a variety of bottom-up estimates to be consistent with atmospheric CH₄ observations. Global total microbial CH₄ emissions, including emissions from natural wetlands, lakes, reservoirs and wild animals, and anthropogenic sources of livestock, waste/landfills, and rice cultivation, are further constrained by the stable carbon isotope of CH₄ (delta⁻¹³CH₄), suggesting a total emission ranging from 360 to 388 Tg CH₄ yr⁻¹ for 2000–2009 (Lan et al., 2021). Wetland emissions from the top-down CH₄ inversions are estimated to be in the range of 153–196 CH₄ yr⁻¹, while anthropogenic microbial emissions are in the range of 198–219 CH₄ yr⁻¹ for 2000–2009. If we assume other sink and source estimates are accurate, the lake and reservoirs emissions need to be small to match the atmospheric burden constrained by CH₄ and delta-¹³CH₄ observations. The large estimates of 70.9 Tg CH₄ yr⁻¹ by Rosentreter et al. (2021) and 122–159 Tg CH₄ yr⁻¹ by Saunois et al. (2020) for global freshwater methane emissions will need to have larger sink terms or smaller source terms in the global methane cycling to reconcile the atmospheric observations and modeled concentrations.

3.3. Warming Effects Under Future Scenarios

Future warming enhances microbial activities, contributing to increased methane emissions (Guo et al., 2020; Yvon-Durocher et al., 2014; Zhu et al., 2020). The estimated total emissions by the end of the 21st century are $44.6 \pm 15.1 \text{ Tg CH}_4 \text{ yr}^{-1}$ and $37.8 \pm 11.2 \text{ Tg CH}_4 \text{ yr}^{-1}$, increases by 86% and 58% compared to the 2000s, under the RCP8.5 using climate outputs from HadGEM2-ES and GFDL-ESM2M, respectively (Table 2 and Figure 3). Under both warming conditions, the pan-Arctic has the fastest increase, over 160% within the century, followed by temperate (20%–30%) and tropical regions (3%–20%). The ratio of diffusive to ebullitive emissions decreases by 10% and 5% using HadGEM2-ES and GFDL-ESM2M, respectively. Ebullition is more sensitive to warming, as suggested in Aben et al. (2017), but underlying causes are uncertain.

Our estimated increase in global lake emissions tends to be high. A recent projection suggested that future warming of bottom waters will likely increase methane production by 13%–40% by the end of the century (Jansen et al., 2022), but they also indicated that many low-latitude lakes experience an increase of up to 17 times the historical (1970–1999) global average under RCP8.5. For the pan-Arctic, our estimate of total emissions is lower than 28.1 Tg CH₄ yr⁻¹ for the region north of 60°N by Tan et al. (2015). The higher estimate by Tan et al. (2015) is probably because they calibrated the model against five lakes, four of which are small thermokarst lakes whereas the less productive large glacial lakes cover most area in this region (Matveev et al., 2016; Walter Anthony et al., 2016; Wik et al., 2016). We predict a higher emission increase than the increase of 29% estimated by Wik et al. (2016). Our previous study suggested that ice-free days play a role in rising methane emissions from boreal lakes (Guo et al., 2020). Furthermore, an increase from enhanced microbial production is included in our estimate. This reflects the importance of explicit representation of lake thermal and carbon dynamics for future projections.





Figure 3. Simulations under RCP8.5 using HadGEM2-ES and GFDL-ESM2M for mean annual lake methane fluxes averaged for the two models (a). The latitudinal distributions of land air temperature increase in the 2090s compared to the 2000s (b), annual total methane emissions (c), and annual methane fluxes averaged by lake surface area (d).

In the model, the average summer (July, August, and September—JAS) methane production rate increased by 19% and 36% for the pan-Arctic, 25% and 40% for the temperate, and 8% and 22% for the tropics using GFDL-ESM2M and HadGEM2-ES, respectively. It has been found that methanogens are more climate sensitive than methanotrophs (Zhu et al., 2020), which is also shown in the calibrated PQ_{10} and OQ_{10} values which range within 1.7–5.5 and 1.0–1.2, respectively. However, the simulations show the opposite for the temperate and the tropics (Table 2). Based on modeled process rates, methane oxidation increased at a comparable rate to methane production for most of the lakes in these two regions (Figure 4). Based on Equation 3, this may be caused by higher dissolved oxygen concentrations, higher methane concentrations, or both. The oxygen concentration is likely to decrease because solubility decreases with increasing water temperature. Also, according to Woolway and Merchant (2019), global lake mixing regimes are shifting toward stronger and longer stratification under warming, leading to weaker mixing of oxygen to lower depths. Thus, increase in water column oxygen is unlikely the cause of higher methane oxidation rates while increasing methane concentration resulting from enhanced







Figure 4. Ratios of summer (JAS) lake sediment methane production rate (PCH_4) and summer methane oxidation rates modeled in surface ($OCH_{4,surf}$), middle($OCH_{4,mid}$), and bottom ($OCH_{4,bot}$) layers of the water under RCP8.5 (rate_{rcp8.5}) to the current rate in the pan-Arctic (a), temperate (b), and tropics (c). The middle layer is simply the middle point between the surface and bottom.

production is more likely the major cause. Our modeling result is consistent with several experimental studies that measured process rates and feedbacks under controlled warming and found methanotrophy is enhanced similarly to increased methanogenesis (Lofton et al., 2014; Shelley et al., 2015).

3.4. Uncertainties in Permafrost Dynamics Impacts

The doubling in SOC results in a 56% increase of 5.5 Tg CH_4 yr⁻¹ in the total methane emissions. Diffusive and ebullitive fluxes account for 47% and 57% of the increase, respectively. By analyzing the increases of methane oxidation versus production rates with doubled SOC (same as Figure 4, not shown), we find a consistent pattern with the warming-induced methanogenesis increase. The extra SOC input by thawing permafrost is not fully translated to methane emission to the atmosphere, which again demonstrates that despite the growing methane source due to warming and old carbon input, methane oxidation in lake water can serve as an important sink that counteracts part of the climate impacts. In this scenario, the lake classification remained the same as the present because landscape evolution is not within the scope of this study.

Our modeled current total methane emission from thermokarst lakes larger than 0.1 km² is 2.3 ± 0.7 Tg CH₄ yr⁻¹. Wik et al. (2016) gave an estimate of 4.1 ± 2.2 Tg CH₄ yr⁻¹ from thermokarst lakes larger than 0.002 km² using statistical upscaling. If the difference between these results is due to lakes smaller than 0.1 km² with the same number of small lakes from these studies, the total emissions from thermokarst lakes becomes 2.9 Tg CH₄ yr⁻¹. With the thermokarst lake area changing by $\pm 30\%$, a change of ± 1.65 Tg CH₄ yr⁻¹ is calculated from the modeling. If lakes smaller than 0.1 km² are included, the total variation would be ± 2.1 Tg CH₄ yr⁻¹, about 28% and 9% of the current pan-Arctic and global total, respectively. Therefore, the thermokarst lake areal change can have an impact at the regional scale but has a small role in global lake methane emissions.

3.5. Research Limitations

There are several limitations in the study. First, difficulties in characterizing lake properties introduce uncertainties to our simulations. Classifying all lakes without water quality information as oligotrophic is likely to underestimate emissions. Sediment carbon stocks are heterogeneous, and a full representation would involve local carbon



inputs and outputs, and simulation of the full carbon cycle that includes algae, plants and microbial activities, and carbon sedimentation and burial. These processes are seldom fully quantified, and parameter values are poorly constrained due to lack of measurements. Although the model performed well in both calibration and validation (Figure S2 in Supporting Information S1), using parameters derived from current states to make future projections is based on the assumption of parameter stationarity. Riley et al. (2011) noted this as a universal issue with prediction by process-based modeling. Furthermore, several algorithms, such as those used to model ebullition, gas exchange at the air-water interface and vertical mixing, can be improved.

Second, our predictions are significantly limited by the quality and quantity of observational data for diverse lakes. This uncertainty could be from our calibration for various processes of lake methane cycling including methane production, oxidation, and transport pathways. In addition, our calibration and verification are conducted primarily for individual lakes smaller than 20 km² with a few exceptions, while the global lake area includes large lakes. This will likely overestimate our regional and global simulations when applying parameters obtained from relatively small lakes to large lakes. Future analysis shall be based on more emission data for large lakes when they are available.

Third, lacking regional-scale measurements of permafrost thaw impacts on pan-Arctic lakes is a major uncertainty in predicting lake methane emissions by the end of the century, especially when over 90% of the permafrost is estimated to disappear under RCP8.5 by multiple climate models (Koven et al., 2013). Previous studies estimating the amount of thawed old carbon can shed lights on the impacts on lakes (Hugelius et al., 2020; Shurr et al., 2015), but to incorporate the process into models, lake-specific projections are needed. Site-level measurements have shown diverse changes in lake SOC input (Walter Anthony et al., 2016) and thus, it requires further understanding from the mechanistic aspect to gain more insights into the issue. Our sensitivity analysis with the doubled SOC input to lakes is a relatively simple treatment. More detailed data on lateral flow of SOC from land to aquatic ecosystems considering permafrost thaw and thermokarst will improve our future lake methane emission analysis.

Another important uncertainty is the global lake distribution data, especially for small lakes (<0.1 km²). These small lakes could be regenerated, expanded, or disappeared due to thawing permafrost and thermokarst in the Arctic and their emission rates are large. For instance, Johnson et al. (2021) using a new spatially explicit data set of lakes >50°N, included lakes with areas smaller than 0.1 km². Matthews et al. (2020) estimated that the area of lakes north of 50°N is 10.95×10^5 km². while our estimates are based on HydroLAKES data with lake and water reservoir area of 2.49 × 10⁵ km² only. Thus, including small lakers' areal dynamics should improve our future regional estimates.

Finally, the water column methanotrophs are simulated for three water layers and their significant role in affecting the lake methane emissions are modeled. However, the methanotroph parameterization is compromised because it is not based on observed methane oxidation rate, rather based on the net lake emission data. Methane oxidation rates could vary significantly from 0.25 to 81 mg of C $m^{-2} day^{-1}$ for different lakes (Bastviken et al., 2002). Additionally, oxidation rates at different lake depths within the same lake could also vary (Langenegger et al., 2022). Thus, parameterizing the model with depth oxidation profiles for various lakes will constrain our model uncertainty.

4. Conclusions

We estimate that global methane emissions from lakes and reservoirs larger than 0.1 km² using an improved modeling approach are 24.0 \pm 8.4 Tg CH₄ yr⁻¹ for the present, which is lower than previous statistical upscaling results. Under the RCP8.5 warming scenario, our sensitivity analysis suggests the global total emissions will increase by 58%–86% with an assumption that the carbon substrate for lake methanogenesis is doubled. We further investigate the emission sensitivity to different factors, including warming, carbon input from thawed permafrost, and changes in thermokarst lake areas, on lake methane emissions under changing climate conditions. We find that, although carbon from thawed permafrost can increase methane production, methane oxidation would also be enhanced due to increasing methane concentration. Our hypothetical analysis on thermokarst areal changes, either shrinking or expanding, suggests they exert a minor impact on global total methane emissions. Our study suggests that explicitly incorporating processes related to lake thermal and carbon dynamics (e.g., lateral flow of carbon from land to aquatic ecosystems) is important to future projections while developing and modeling lake distribution dynamics is a research priority.



Data Availability Statement

Research data generated in this study are available via doi: https://doi.org/10.4231/JZ10-FH54 and the model code is available via doi: https://doi.org/10.4231/67YG-V518 in Purdue University Research Repository.

References

- Aben, R. C., Barros, N., Van Donk, E., Frenken, T., Hilt, S., Kazanjian, G., et al. (2017). Cross continental increase in methane ebullition under climate change. *Nature communications*, 8, 1682. https://doi.org/10.1038/s41467-017-01535-y
- Amaral, J. H. F., Melack, J. M., Maia Barbosa, P., MacIntyre, S., Kasper, D., Cortés, A., et al. (2020). Carbon dioxide fluxes to the atmosphere from waters within flooded forests in the Amazon basin. *Journal of Geophysical Research: Biogeosciences*, 125, e2019JG005293. https://doi. org/10.1029/2019JG005293
- Attermeyer, K., Flury, S., Jayakumar, R., Fiener, P., Steger, K., Arya, V., et al. (2016). Invasive floating macrophytes reduce greenhouse gas emissions from a small tropical lake. *Scientific Reports*, 6(1), 20424. https://doi.org/10.1038/srep20424
- Bastviken, D., Ejlertsson, J., & Tranvik, L. (2002). Measurement of methane oxidation in lakes: A comparison of methods. *Environmental Science & Technology*, 36(15), 3354–3361. https://doi.org/10.1021/es010311p
- Brown, J., Ferrians, O. J., Jr., Heginbottom, J. A., & Melnikov, E. S. (1997). Circum-Arctic map of permafrost and ground-ice conditions. *Circum-Pacific Map*, 45. https://doi.org/10.3133/cp45
- Deemer, B. R., Harrison, J. A., Li, S., Beaulieu, J. J., DelSontro, T., Barros, N., et al. (2016). Greenhouse gas emissions from reservoir water surfaces: A new global synthesis. *BioScience*, 66(11), 949–964. https://doi.org/10.1093/biosci/biw117
- Filazzola, A., Mahdiyan, O., Shuvo, A., Ewins, C., Moslenko, L., Sadid, T., et al. (2020). A database of chlorophyll and water chemistry in freshwater lakes. *Scientific Data*, 7(1), 310. https://doi.org/10.1038/s41597-020-00648-2
- Frieler, K., Lange, S., Piontek, F., Reyer, C. P. O., Schewe, J., Warszawski, L., et al. (2017). Assessing the impacts of 1.5°C global warming— Simulation protocol of the Inter-Sectoral Impact Model Intercomparison Project (ISIMIP2b). *Geoscientific Model Development*, 10(12), 4321–4345. https://doi.org/10.5194/gmd-10-4321-2017
- Fung, I., John, J., Lerner, J., Matthews, E., Prather, M., Steele, L. P., & Fraser, P. J. (1991). Three-dimensional model synthesis of the global methane cycle. *Journal of Geophysical Research*, 96(D7), 13033–13065. https://doi.org/10.1029/91JD01247
- Guo, M., Zhuang, Q., Tan, Z., Shurpali, N., Juutinen, S., Kortelainen, P., & Martikainen, P. J. (2020). Rising methane emissions from boreal lakes due to increasing ice-free days. *Environmental Research Letters*, 15(6), 064008. https://doi.org/10.1088/1748-9326/ab8254
- Guo, M., Zhuang, Q., Yao, H., Golub, M., Leung, L. R., Pierson, D., & Tan, Z. (2021). Validation and sensitivity analysis of a 1-D lake model across global lakes. *Journal of Geophysical Research: Atmospheres*, 126, e2020JD033417. https://doi.org/10.1029/2020JD033417
- Guo, M., Zhuang, Q., Yao, H., Golub, M., Leung, L. R., & Tan, Z. (2021). Intercomparison of thermal regime algorithms in 1-D lake models. Water Resources Research, 57, e2020WR028776. https://doi.org/10.1029/2020WR028776
- Guseva, S., Bleninger, T., Jöhnk, K., Polli, B. A., Tan, Z., Thiery, W., et al. (2020). Multimodel simulation of vertical gas transfer in a temperate lake. *Hydrology Journal of Earth System Science*, 24(2), 697–715. https://doi.org/10.5194/hess-24-697-2020
- Hartmann, J., Lauerwald, R., & Moosdorf, N. (2014). A brief overview of the GLObal RIver CHemistry Database, GLORICH. Procedia Earth and Planetary Science, 10, 23–27. https://doi.org/10.1016/j.proeps.2014.08.005
- Heiskanen, J. J., Mammarella, I., Haapanala, S., Pumpanen, J., Vesala, T., Macintyre, S., & Ojala, A. (2014). Effects of cooling and internal wave motions on gas transfer coefficients in a boreal lake. *Tellus Series B: Chemical and Physical Meteorology*, 66(1), 22827. https://doi. org/10.3402/tellu
- Hugelius, G., Loisel, J., Chadburn, S., Jackson, R. B., Jones, M., MacDonald, G., et al. (2020). Large stocks of peatland carbon and nitrogen are vulnerable to permafrost thaw. *Proceedings of the National Academy of Sciences of the United States of America*, 117(34), 20438–20446. https://doi.org/10.1073/pnas.1916387117
- Jansen, J., Woolway, R. I., Kraemer, B. M., Albergel, C., Bastviken, D., Weyhenmeyer, G. A., et al. (2022). Global increase in methane production under future warming of lake bottom waters. *Global Change Biology*, 28(18), 5427–5440. https://doi.org/10.1111/gcb.16298
- Johnson, M. S., Matthews, E., Bastviken, D., Deemer, B., Du, J., & Genovese, V. (2021). Spatiotemporal methane emission from global reservoirs. *Journal of Geophysical Research: Biogeosciences*, 126, e2021JG006305. https://doi.org/10.1029/2021JG006305
- Johnson, M. S., Matthews, E., Du, J., Genovese, V., & Bastviken, D. (2022). Methane emission from global lakes: New spatiotemporal data and observation-driven modeling of methane dynamics indicates lower emissions. *Journal of Geophysical Research: Biogeosciences*, 127, e2022JG006793. https://doi.org/10.1029/2022JG006793
- Jones, B. M., Grosse, G. D. A. C., Arp, C. D., Jones, M. C., Anthony, K. W., & Romanovsky, V. E. (2011). Modern thermokarst lake dynamics in the continuous permafrost zone, northern Seward Peninsula, Alaska. *Journal of Geophysical Research*, 116, G00M03. https://doi. org/10.1029/2011JG001666
- Keaveney, E. M., Radbourne, A. D., McGowan, S., Ryves, D. B., & Reimer, P. J. (2020). Source and quantity of carbon influence its sequestration in Rostherne Mere (UK) sediment: A novel application of stepped combustion radiocarbon analysis. *Journal of Paleolimnology*, 64(4), 347–363. https://doi.org/10.1007/s10933-020-00141-1
- Kessler, M. A., Plug, L. J., & Walter Anthony, K. M. (2012). Simulating the decadal-to millennial-scale dynamics of morphology and sequestered carbon mobilization of two thermokarst lakes in NW Alaska. *Journal of Geophysical Research*, 117, G00M06. https://doi. org/10.1029/2011JG001796
- Klump, J. V., Paddock, R., Remsen, C. C., Fitzgerald, S., Boraas, M., & Anderson, P. (1989). Variations in sediment accumulation rates and the flux of labile organic matter in eastern Lake Superior basins. *Journal of Great Lakes Research*, 15(1), 104–122. https://doi.org/10.1016/s0380-1330(89)71465-9
- Koven, C. D., Lawrence, D. M., & Riley, W. J. (2015). Permafrost carbon-climate feedback is sensitive to deep soil carbon decomposability but not deep soil nitrogen dynamics. *Proceedings of the National Academy of Sciences of the United States of America*, 112(12), 3752–3757. https://doi.org/10.1073/pnas.1415123112
- Koven, C. D., Riley, W. J., & Stern, A. (2013). Analysis of permafrost thermal dynamics and response to climate change in the CMIP5 Earth System Models. *Journal of Climate*, 26(6), 1877–1900. https://doi.org/10.1175/jcli-d-12-00228.1
- Lan, X., Basu, S., Schwietzke, S., Bruhwiler, L. M. P., Dlugokencky, E. J., Michel, S. E., et al. (2021). Improved constraints on global methane emissions and sinks using δ13C-CH4. *Global Biogeochemical Cycles*, *35*, e2021GB007000. https://doi.org/10.1029/2021GB007000
- Lange, S. (2019). Trend-preserving bias adjustment and statistical downscaling with ISIMIP3BASD (v1.0). Geoscientific Model Development, 12(7), 3055–3070. https://doi.org/10.5194/gmd-12-3055-2019

Acknowledgments

This study was supported through a project funded to O.Z. by NASA (NNX17AK20G) and a project from the United States Geological Survey (G17AC00276). JMM received support from National Aeronautics and Space Administration (NASA Grant NNX17AK49G) and the US National Science Foundation (NSF DEB Grant 1753856). ZT and LRL were supported by the Office of Science of the US Department of Energy as part of the Earth System Model Development program area through the Energy Exascale Earth System Model (E3SM) project. The Pacific Northwest National Laboratory was operated by Battelle for the US Department of Energy under Contract DE-AC05-76RLO1830.



- Langenegger, T., Vachon, D., Donis, D., & McGinnis, D. F. (2022). Methane oxidation dynamics in a stratified lake: Insights revealed from a mass balance and carbon stable isotopes. *Limnology & Oceanography*, 67(10), 2157–2173. https://doi.org/10.1002/lno.12195
- Liang, J. H., McWilliams, J. C., Sullivan, P. P., & Baschek, B. (2011). Modeling bubbles and dissolved gases in the ocean. Journal of Geophysical Research, 116, C03015. https://doi.org/10.1029/2010JC006579
- Liikanen, A., Huttunen, J. T., Valli, K., & Martikainen, P. J. (2002). Methane cycling in the sediment and water column of mid-boreal hyper-eutrophic Lake Kevätön, Finland. Archiv fur Hydrobiologie, 154(4), 585–603. https://doi.org/10.1127/archiv-hydrobiol/154/2002/585
- Lofton, D. D., Whalen, S. C., & Hershey, A. E. (2014). Effect of temperature on methane dynamics and evaluation of methane oxidation kinetics in shallow Arctic Alaskan lakes. *Hydrobiologia*, 721(1), 209–222. https://doi.org/10.1007/s10750-013-1663-x
- Marani, L., & Alvalá, P. C. (2007). Methane emissions from lakes and floodplains in Pantanal, Brazil. Atmospheric Environment, 41(8), 1627– 1633. https://doi.org/10.1016/j.atmosenv.2006.10.046
- Matthews, E., Johnson, M. S., Genovese, V., Du, J., & Bastviken, D. (2020). Methane emission from high latitude lakes: Methane-centric lake classification and satellite-driven annual cycle of emissions. *Scientific Reports*, 10(1), 12465. https://doi.org/10.1038/s41598-020-68246-1
- Matveev, A., Laurion, I., Deshpande, B. N., Bhiry, N., & Vincent, W. F. (2016). High methane emissions from thermokarst lakes in subarctic peatlands. *Limnology & Oceanography*, 61(S1), S150–S164. https://doi.org/10.1002/ino.10311
- Melton, J. R., Wania, R., Hodson, E. L., Poulter, B., Ringeval, B., Spahni, R., et al. (2013). Present state of global wetland extent and wetland methane modelling: Conclusions from a model inter-comparison project (WETCHIMP). *Biogeosciences*, 10(2), 753–788. https://doi. org/10.5194/bg-10-753-2013
- Messager, M. L., Lehner, B., Grill, G., Nedeva, I., & Schmitt, O. (2016). Estimating the volume and age of water stored in global lakes using a geo-statistical approach. *Nature Communications*, 7(1), 13603. https://doi.org/10.1038/ncomms13603
- Nitze, I., Cooley, S. W., Duguay, C. R., Jones, B. M., & Grosse, G. (2020). The catastrophic thermokarst lake drainage events of 2018 in northwestern Alaska: Fast-forward into the future. *The Cryosphere*, 14(12), 4279–4297. https://doi.org/10.5194/tc-14-4279-2020
- OECD (Organisation for Economic Co-operation and Development). (1992). Report of the OECD workshop on the extrapolation of laboratory aquatic toxicity data to the real environment. Arlington, Virginia, December 1990. OECD Environment Monograph 59.
- Olefeldt, D., Goswami, S., Grosse, G., Hayes, D., Hugelius, G., Kuhry, P., et al. (2016). Circumpolar distribution and carbon storage of thermokarst landscapes. *Nature Communications*, 7(1), 13043. https://doi.org/10.1038/ncomms13043
- Ostapenia, A. P., Parparov, A., & Berman, T. (2009). Lability of organic carbon in lakes of different trophic status. *Freshwater Biology*, 54(6), 1312–1323. https://doi.org/10.1111/j.1365-2427.2009.02183.x
- Remsen, C. C., Minnich, E. C., Stephens, R. S., Buchholz, L., & Lidstrom, M. E. (1989). Methane oxidation in lake superior sediments. Journal of Great Lakes Research, 15(1), 141–146. https://doi.org/10.1016/s0380-1330(89)71468-4
- Riley, W. J., Subin, Z. M., Lawrence, D. M., Swenson, S. C., Torn, M. S., Meng, L., et al. (2011). Barriers to predicting changes in global terrestrial methane fluxes: Analyses using CLM4Me, a methane biogeochemistry model integrated in CESM. *Biogeosciences*, 8(7), 1925–1953. https://doi.org/10.5194/bg-8-1925-2011
- Rosentreter, J. A., Borges, A. V., Deemer, B. R., Holgerson, M. A., Liu, S., Song, C., et al. (2021). Half of global methane emissions come from highly variable aquatic ecosystem sources. *Nature Geoscience*, 14(4), 225–230. https://doi.org/10.1038/s41561-021-00715-2
- Saunois, M., Bousquet, P., Poulter, B., Peregon, A., Ciais, P., Canadell, J. G., et al. (2016). The global methane budget 2000–2012. Earth System Science Data, 8(2), 697–751. https://doi.org/10.5194/essd-8-697-2016
- Saunois, M., Stavert, A. R., Poulter, B., Bousquet, P., Canadell, J. G., Jackson, R. B., et al. (2020). The global methane budget 2000–2017. Earth System Science Data, 12(3), 1561–1623. https://doi.org/10.5194/essd-12-1561-2020
- Schädel, C., Schuur, E. A., Bracho, R., Elberling, B., Knoblauch, C., Lee, H., et al. (2014). Circumpolar assessment of permafrost C quality and its vulnerability over time using long-term incubation data. *Global Change Biology*, 20(2), 641–652. https://doi.org/10.1111/gcb.12417
- Schuur, E. A., Vogel, J. G., Crummer, K. G., Lee, H., Sickman, J. O., & Osterkamp, T. E. (2009). The effect of permafrost thaw on old carbon release and net carbon exchange from tundra. *Nature*, 459(7246), 556–559. https://doi.org/10.1038/nature08031
- Schuur, E., McGuire, A., Schädel, C., Grosse, G., Harden, J. W., Hayes, D. J., et al. (2015). Climate change and the permafrost carbon feedback. *Nature*, 520(7546), 171–179. https://doi.org/10.1038/nature14338
- Segers, R. (1998). Methane production and methane consumption: A review of processes underlying wetland methane fluxes. *Biogeochemistry*, 41(1), 23–51.
- Serikova, S., Pokrovsky, O. S., Laudon, H., Krickov, I. V., Lim, A. G., Manasypov, R. M., & Karlsson, J. (2019). High carbon emissions from thermokarst lakes of Western Siberia. *Nature Communications*, 10(1), 1552. https://doi.org/10.1038/s41467-019-09592-1
- Shelley, F., Abdullahi, F., Grey, J., & Trimmer, M. (2015). Microbial methane cycling in the bed of a chalk river: Oxidation has the potential to match methanogenesis enhanced by warming. *Freshwater Biology*, 60(1), 150–160. https://doi.org/10.1111/fwb.12480
- Strauss, J., Laboor, S., Fedorov, A. N., Fortier, D., Froese, D. G., Fuchs, M., et al. (2016). Database of Ice-Rich Yedoma Permafrost (IRYP). *PANGAEA*. https://doi.org/10.1594/PANGAEA.861733
- Tan, Z., Yao, H., & Zhuang, Q. (2018). A small temperate lake in the 21st century: Dynamics of water temperature, ice phenology, dissolved oxygen, and chlorophyll A. Water Resources Research, 54, 4681–4699. https://doi.org/10.1029/2017WR022334
- Tan, Z., & Zhuang, Q. (2015). Arctic lakes are continuous methane sources to the atmosphere under warming conditions. Environmental Research Letters, 10(5), 054016. https://doi.org/10.1088/1748-9326/10/5/054016
- Tan, Z., Zhuang, Q., Shurpali, N. J., Marushchak, M. E., Biasi, C., Eugster, W., & Anthony, K. W. (2017). Modeling CO₂ emissions from Arctic lakes: Model development and site-level study. *Journal of Advances in Modeling Earth Systems*, 9, 2190–2213. https://doi. org/10.1002/2017MS001028
- Tan, Z., Zhuang, Q., & Walter Anthony, K. (2015). Modeling methane emissions from arctic lakes: Model development and site-level study. Journal of Advances in Modeling Earth Systems, 7, 459–483. https://doi.org/10.1002/2014MS000344
- Tang, J., & Zhuang, Q. (2009). A global sensitivity analysis and Bayesian inference framework for improving the parameter estimation and prediction of a process-based Terrestrial Ecosystem Model. Journal of Geophysical Research, 114, D15303. https://doi.org/10.1029/2009JD011724
- Turetsky, M. R., Abbott, B. W., Jones, M. C., Walter Anthony, K., Olefeldt, D., Schuur, E. A. G., et al. (2020). Carbon release through abrupt permafrost thaw. *Nature Geoscience*, 13(2), 138–143. https://doi.org/10.1038/s41561-019-0526-0
- van Bodegom, P. M., Wassmann, R., & Metra-Corton, T. M. (2001). A process-based model for methane emission predictions from flooded rice paddies. *Global Biogeochemical Cycles*, 15(1), 247–263. https://doi.org/10.1029/1999GB001222
- Van Huissteden, J., Berrittella, C., Parmentier, F. J. W., Mi, Y., Maximov, T. C., & Dolman, A. J. (2011). Methane emissions from permafrost thaw lakes limited by lake drainage. *Nature Climate Change*, 1(2), 119–123. https://doi.org/10.1038/nclimate1101
- Verpoorter, C., Kutser, T., Seekell, D. A., & Tranvik, L. J. (2014). A global inventory of lakes based on high-resolution satellite imagery. Geophysical Research Letters, 41, 6396–6402. https://doi.org/10.1002/2014GL060641



- von Deimling, S. T., Grosse, G., Strauss, J., Schirrmeister, L., Morgenstern, A., Schaphoff, S., et al. (2015). Observation-based modelling of permafrost carbon fluxes with accounting for deep carbon deposits and thermokarst activity. *Biogeosciences*, 12(11), 3469–3488. https://doi. org/10.5194/bg-12-3469-2015
- Walter, B. P., & Heimann, M. (2000). A process-based, climate-sensitive model to derive methane emissions from natural wetlands: Application to five wetland sites, sensitivity to model parameters, and climate. *Global Biogeochemical Cycles*, 14(3), 745–765. https://doi. org/10.1029/1999GB001204
- Walter, K. M., Smith, L. C., & Stuart Chapin, F., III. (2007). Methane bubbling from northern lakes: Present and future contributions to the global methane budget. *Philosophical Transactions of the Royal Society A: Mathematical, Physical & Engineering Sciences*, 365(1856), 1657–1676. https://doi.org/10.1098/rsta.2007.2036
- Walter Anthony, K., Daanen, R., Anthony, P., von Deimling, T. S., Ping, C.-L., Chanton, J. P., & Grosse, G. (2016). Methane emissions proportional to permafrost carbon thawed in Arctic lakes since the 1950s. *Nature Geoscience*, 9, 679–682. https://doi.org/10.1038/ngeo2795
- Walter Anthony, K., Schneider von Deimling, T., Nitze, I., Frolking, S., Emond, A., Daanen, R., et al. (2018). 21st-century modeled permafrost carbon emissions accelerated by abrupt thaw beneath lakes. *Nature Communications*, 9(1), 3262. https://doi.org/10.1038/s41467-018-05738-9
 Walter Anthony, K. M., Anthony, P., Grosse, G., & Chanton, J. (2012). Geologic methane seeps along boundaries of Arctic permafrost thaw and
- melting glaciers. Nature Geoscience, 5(6), 419–426. https://doi.org/10.1038/ngeo1480
 Wik, M., Varner, R. K., Anthony, K. W., MacIntyre, S., & Bastviken, D. (2016). Climate-sensitive northern lakes and ponds are critical components of methane release. Nature Geoscience, 9(2), 99–105. https://doi.org/10.1038/ngeo2578
- Woolf, D. K., & Thorpe, S. A. (1991). Bubbles and the air-sea exchange of gases in near-saturation conditions. *Journal of Marine Research*, 49(3), 435–466. https://doi.org/10.1357/002224091784995765
- Woolway, R. I., & Merchant, C. J. (2019). Worldwide alteration of lake mixing regimes in response to climate change. Nature Geoscience, 12(4), 271–276. https://doi.org/10.1038/s41561-019-0322-x
- Yukimoto, S., Adachi, Y., Hosaka, M., Sakami, T., Yoshimura, H., Hirabara, M., et al. (2012). A new global climate model of the meteorological research institute: MRI-CGCM3—Model description and basic performance. *Journal of the Meteorological Society of Japan*, 90A, 23–64. https://doi.org/10.2151/jmsj.2012-A02
- Yvon-Durocher, G., Allen, A. P., Bastviken, D., Conrad, R., Gudasz, C., St-Pierre, A., et al. (2014). Methane fluxes show consistent temperature dependence across microbial to ecosystem scales. *Nature*, 507(7493), 488–491. https://doi.org/10.1038/nature13164
- Zhu, Y., Purdy, K. J., Eyice, Ö., Shen, L., Harpenslager, S. F., Yvon-Durocher, G., et al. (2020). Disproportionate increase in freshwater methane emissions induced by experimental warming. *Nature Climate Change*, *10*(7), 685–690. https://doi.org/10.1038/s41558-020-0824-y
- Zhuang, Q., Melillo, J. M., Kicklighter, D. W., Prinn, R. G., McGuire, A. D., Steudler, P. A., et al. (2004). Methane fluxes between terrestrial ecosystems and the atmosphere at northern high latitudes during the past century: A retrospective analysis with a process-based biogeochemistry model. *Global Biogeochemical Cycles*, *18*, GB3010. https://doi.org/10.1029/2004GB002239